# Long-Lived Antibody and B Cell Memory Responses to the Human Malaria Parasites, *Plasmodium falciparum* and *Plasmodium vivax*

Jiraprapa Wipasa<sup>1</sup>, Chaisuree Suphavilai<sup>1</sup>, Lucy C. Okell<sup>2</sup>, Jackie Cook<sup>2</sup>, Patrick H. Corran<sup>2</sup>, Kanitta Thaikla<sup>1</sup>, Witaya Liewsaree<sup>3</sup>, Eleanor M. Riley<sup>2</sup>\*, Julius Clemence R. Hafalla<sup>2</sup>\*

1 Research Institute for Health Sciences, Chiang Mai University, Chiang Mai, Thailand, 2 Department of Infectious and Tropical Diseases, London School of Hygiene and Tropical Medicine, London, United Kingdom, 3 Vector Borne Disease Section, Office of Disease Prevention and Control, Chiang Mai, Thailand

# Abstract

Antibodies constitute a critical component of the naturally acquired immunity that develops following frequent exposure to malaria. However, specific antibody titres have been reported to decline rapidly in the absence of reinfection, supporting the widely perceived notion that malaria infections fail to induce durable immunological memory responses. Currently, direct evidence for the presence or absence of immune memory to malaria is limited. In this study, we analysed the longevity of both antibody and B cell memory responses to malaria antigens among individuals who were living in an area of extremely low malaria transmission in northern Thailand, and who were known either to be malaria naïve or to have had a documented clinical attack of *P. falciparum* and/or *P. vivax* in the past 6 years. We found that exposure to malaria results in the generation of relatively avid antigen-specific antibodies and the establishment of populations of antigen-specific memory B cells in a significant proportion of malaria-exposed individuals. Both antibody and memory B cell responses to malaria antigens were stably maintained over time in the absence of reinfection. In a number of cases where antigen-specific antibodies were not detected in plasma, stable frequencies of antigen-specific memory B cells were nonetheless observed, suggesting that circulating memory B cells may be maintained independently of long-lived plasma cells. We conclude that infrequent malaria infections are capable of inducing long-lived antibody and memory B cell responses.

Citation: Wipasa J, Suphavilai C, Okell LC, Cook J, Corran PH, et al. (2010) Long-Lived Antibody and B Cell Memory Responses to the Human Malaria Parasites, *Plasmodium falciparum* and *Plasmodium vivax*. PLoS Pathog 6(2): e1000770. doi:10.1371/journal.ppat.1000770

Editor: James W. Kazura, Case Western Reserve University, United States of America

Received June 12, 2009; Accepted January 14, 2010; Published February 19, 2010

**Copyright:** © 2010 Wipasa et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

**Funding:** This study was funded by a Wellcome Trust International Research Development Award (grant reference number 072182; to JW), a Wellcome Trust Visiting Fellowship (grant reference number 079920; to JCRH) and a Royal Society Incoming Fellowship (to JCRH). The Wellcome Trust (www.wellcome.ac.uk) and The Royal Society (www.royalsoc.ac.uk) had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

\* E-mail: eleanor.riley@lshtm.ac.uk (EMR); Julius.Hafalla@lshtm.ac.uk (JCRH)

# Introduction

Malaria, a parasitic disease of humans caused predominantly by two species of Plasmodium, P. falciparum and P. vivax, remains an important cause of mortality and morbidity in many parts of the world. Development of a vaccine against malaria has proven challenging due to the complex nature of the parasite and to the difficulty in correlating naturally-acquired immune responses with clinical immunity. While immunity against some of the severe clinical symptoms may be achieved quite rapidly, following perhaps as few as one or two infections [1], immune effector mechanisms capable of controlling parasite growth develop only after repeated infections over a number of years. Even with repeated infections, protective immunity to malaria is not complete, and asymptomatic infections may exist throughout life. Understanding the causes of this continuing susceptibility to infection and, in particular, understanding the development and maintenance of immunological memory, is essential for rational development of malaria vaccines.

Antibodies are a crucial component of naturally acquired protective immunity against blood stage malaria with roles that may include inhibition of merozoite invasion into new red blood cells (RBCs), blocking cytoadherence of infected RBCs (iRBCs) to endothelial cells, and enhancing phagocytic activity of monocytes and macrophages (reviewed in [2,3]). It is widely believed that periodic reinfection is required to maintain acquired immunity to malaria and that antimalarial antibodies are short-lived in the absence of reinfection (reviewed in [4]); implying that B cell memory to malaria may be defective or suboptimal. However, the development and persistence of B cell memory following malaria infection has long been a matter of debate (reviewed in [5]). Some studies in animal models have shown that memory B cells do develop and are maintained normally after malaria infection [6,7]; whereas others have found that malaria infection interferes with the development of memory B cells and long-lived plasma cells [8,9]. In humans, several studies have demonstrated stable antibody responses to malaria antigens [10,11,12], however, short-lived antibody responses have also been observed [13,14], especially in young children [10,15]. To date, very few studies have examined the induction and maintenance of malaria-specific memory B cells in humans. Dorfman et al [14] were frequently unable to detect circulating malaria-specific B cells in antibody seropositive children, but it is unclear whether this reflects an absence of such cells or a lack of sensitivity in the assays used to

### **Author Summary**

It is widely perceived that immunity to malaria is shortlived, rendering people susceptible to repeated malaria infections. However, there have been very few studies on "memory" responses, how the human immune system recognizes previously encountered malaria parasites. In particular, very little is known about the durability of malaria-specific B cells and antibodies. The aim of this study was to investigate the induction and maintenance of B cell memory responses to malaria parasites in a region of Thailand where people become infected with malaria, but where the levels of malaria transmission are so low that repeated infection is uncommon. From hospital records we were able to identify people who either had been infected with malaria over the past 6 years and/or had never been infected. Blood samples were collected on four separate occasions over a period of one year and analysed by microscopy and PCR for presence of malaria parasites and by ELISA and ELISPOT for anti malarial antibodies and malaria-specific memory B cells. We found that, in a significant proportion of individuals, malaria infection results in the generation of antibodies and the establishment of populations of memory B cells against malaria parasites, which were very stably maintained over time despite the lack of any evidence of malaria reinfection. Contrary to the widely held idea that memory to malaria is suboptimally induced, our data demonstrate that B cell responses to malaria can be maintained for many years after a malaria infection and indicate that there is no inherent reason why malaria vaccines should not also induce long-lasting protection against malaria.

detect them. Conversely, Asito et al [16] observed an increase in both the total  $CD38^+IgD^-$  memory B cell population and the transitional  $CD10^+CD19^+$  B cell population, following an episode of acute malaria in African children but this study lacked any analysis of the specificity of B cell responses as well as any long term follow up to ascertain the duration of the response.

The aim of this study was to investigate the longevity of the human B cell memory response to malaria in individuals with one or more known malaria infections. To do this, we identified individuals living in an area of very low malaria endemicity in Northern Thailand who were either malaria naïve or who had had recorded (and parasitologically confirmed) clinical episodes of *P. falciparum* or *P vivax* infection some years previously and characterised the antibody and memory B cell response to a variety of discrete *P. falciparum* and *P. vivax* antigens under conditions of infrequent re-exposure/boosting of the immune response.

## Results

### Characteristics of the study subjects at recruitment

Malaria-specific humoral immune responses of 93, HIV negative Thai adults were studied (Table 1). Individuals were assigned to one of three groups according to their place of residence and their prior malaria history. Subjects from Chiang Mai were designated "City Naïve" (n = 17). Subjects from Muang Na (Chiang Dao) were designated "Rural with no clinical malaria episode (Rural 1; n = 30)" if they reported no prior episodes of malaria infection and/or if no record of malaria infection was found in the past 6 years.

Muang Na residents who had had one or more fully documented episodes of infection with *P. falciparum*, *P. vivax* or Table 1. Characteristics of study subjects at recruitment.

	City	Rural 1	Rural 2
Total (at recruitment) - no.	17	31	46
Sex - no. (%)			
Male	8 (47.1)	11 (35.5)	25 (54.3)
Female	9 (53.9)	20 (64.5)	21 (46.7)
Age - yr.			
Mean $\pm$ SD	34.7±8.1	32.6±8.5	33.7±7.3
Range	23–46	19–46	19–48
Recorded malaria episodes <sup>[a]</sup> - no	<b>b.</b> (%)		
P. falciparum only			21 (45.7)
P. vivax only			14 (30.4)
Both P. falciparum and P. vivax			6 (13.0)
Unknown			5 (10.9)
Frequency of recorded malaria e	pisodes <sup>[a,b]</sup>	- no.	
Mean P. falciparum $\pm$ SD			$1.25 \pm 0.56$
Range P. falciparum			1–3
Mean P. vivax ± SD			$1.10 \pm 0.26$
Range P. vivax			1–2
Time since last malaria episode <sup>[b</sup>	<sup>1</sup> - mo.		
Mean P. falciparum $\pm$ SD			21.2±12.9
Range P. falciparum			4–58
Mean P. vivax ± SD			20.6±10.1
Range P. vivax			7–39

<sup>[a]</sup>Malaria episodes that were within 45 days apart were considered as a single episode.

<sup>(b)</sup>Malaria episodes based on records from the Office of Vector Borne Disease Control, Department of Communicable Diseases Control, the Ministry of Public Health. Thailand.

doi:10.1371/journal.ppat.1000770.t001

both parasite species, as well as those who recalled a previous infection and were seropositive to P. falciparum schizont extract (PfSE) but for whom hospital records could not be found, were designated as "previously malaria infected" (Rural 2; n = 46). In this group, 21 subjects (45.7%) reported at least one episode of infection with P. falciparum, 14 (30.4%) reported at least one episode of infection with P. vivax and 6 (13.0%) reported infection with both species in the past 6 years. The frequency of malaria infections within the six years prior to recruitment varied from 1-3 episodes (mean  $1.25\pm0.56$  episodes for *P. falciparum* and 1.10±0.26 for P. vivax). Five Rural 2 subjects (10.9%) were strongly seropositive to PfSE and recalled prior malaria episodes, but no documentary evidence of these malaria episodes was found. The time since last documented malaria infections prior to recruitment varied from 4-58 (21.2±12.9) months for those known to have been infected with P. falciparum and 7-39  $(20.6\pm10.1)$  months for those known to have been infected with P. vivax.

Of the 76 rural subjects included at enrolment, 49 (64.5%) were seen again at 3 months, 44 (57.9%) at 6 months and 51 (67.1%) at 12 months. All city individuals were re-sampled 3 months later.

None of the subjects were infected with *P. falciparum* or *P. vivax* - as determined by blood film examination and PCR - at any visit. However, one of 76 rural subjects demonstrated a significant increase in antibody titre during the study (but only to one antigen, PfSE) suggesting that this individual may have experienced a

recent malaria infection, even though they did not report being ill. The three groups did not differ significantly by age or sex.

Antibody levels against tetanus toxoid (TT) were measured by indirect ELISA. There was no significant difference among the groups in the overall levels of antibodies to TT (Fig. 1A) but only 23.7% of the rural subjects (Rural 1 + Rural 2, n = 76) were seropositive for TT at the time of recruitment. Among the rural individuals (Rural 1 + Rural 2) who had blood collected at recruitment and 12 months later (n = 51), the seropositivity rate for TT (19.6%) did not differ between the two time points (Fig. 1G). The individual data for the rural subjects who were seropositive for TT at the time of recruitment and had blood collected more than one time point (n = 17) are shown in Figure 1M.

#### Antibody responses to PfSE

Relative antibody titres to PfSE were measured, at the time of enrolment and at subsequent follow-up. Among the Rural 1 population, 4 (13.3%) individuals had antibody responses above the cut-off, indicating that they had, in fact, been exposed to malaria (Fig. 1B). Among Rural 2 subjects, 25 (54.4%) individuals were seropositive for PfSE. The proportion of seropositives in the Rural 2 group was significantly higher than in the Rural 1 group (p = 0.0003; Fisher's exact test).

There was no difference in the levels of anti-PfSE antibodies between subjects who had been infected with *P. falciparum* only or infected with *P. vivax* only (Fig. S1). Overall, the Rural 2 group had levels of anti-PfSE antibodies that were significantly higher than the Rural 1 group (p = 0.0005; Mann Whitney U test). There was no correlation between the levels of anti-PfSE antibodies and age of the subjects or the number of previous malaria episodes they had experienced (data not shown).

Among the rural subjects (Rural 1 + Rural 2) who had blood collected at the beginning and at the end of the study (n = 51), 19 (37.3%) were seropositive for PfSE at the time of recruitment and all of these remained positive 12 months later (Fig. 1H). The individual data for the rural subjects who were seropositive at the time of recruitment and had blood collected at more than one time point (n = 25) are shown in Figure 1N.

#### Antibody responses to defined *P. falciparum* antigens

Antibody responses to recombinant malaria antigens PfAMA-1, PfMSP-1<sub>19</sub>, PfMSP-2 and PfCSP were examined by indirect ELISA. PfAMA-1, PfMSP-1<sub>19</sub> and PfMSP-2 are antigens of blood stage merozoites. PfCSP is the major surface protein on the surface of sporozoites, the infective stage of the malaria parasite. All of these antigens are key *P. falciparum* vaccine candidates. City naive subjects were seronegative to all *P. falciparum* antigens tested (Fig. 1C–1F). Among the thirty rural individuals with no known episodes of malaria infection in the past 6 years (Rural 1), 7 (23%), 4 (13%), 1 (3%) and 3 (10%) subjects had positive antibody titres against PfAMA-1, PfMSP-1<sub>19</sub>, PfMSP-2 and PfCSP, respectively (Fig. 1C–1F). Seropositivity to individual malaria antigens, and to PfSE, was not significantly correlated (data not shown) and, overall, 10 (33%) Rural 1 subjects were seropositive to one or more *P. falciparum* antigens.

The frequencies of antibody responses to PfAMA-1, PfMSP-1<sub>19</sub>, PfMSP-2 and PfCSP in the previously infected (Rural 2) group were 17 (37%), 22 (48%), 7 (15%) and 4 (9%), respectively. Again, seropositivity to individual malaria antigens was not significantly correlated (data not shown) and, overall, 30 (65%) Rural 2 subjects were seropositive to one or more *P. falciparum* antigens (schizont extract and/or recombinant proteins). The proportion of individuals in the Rural 2 and Rural 1 groups who were seropositive for

recombinant malaria antigens was not significantly different, except that a higher proportion of Rural 2 were seropositive for PfMSP-1<sub>19</sub> (p = 0.003; Fisher's exact test).

The titres of antibodies to  $PfMSP-1_{19}$  and PfMSP-2 were significantly higher among Rural 2 subjects than among Rural 1 subjects but among the Rural 2 group the levels of antibodies to individual *P. falciparum* antigens were not different between previously *P. falciparum* and *P. vivax*- infected subjects (data not shown). Moreover, the titres of antibodies against *P. falciparum* antigens in some Rural 2 subjects were at least as high as those of the positive control of pooled adult African sera. There was no detectable antibody response to the carrier proteins used in the production of recombinant antigens (data not shown).

Several subjects reporting prior infection only with *P. vivax* had antibodies to *P. falciparum* antigens. Of the 14 individuals with recorded *P. vivax* infections but no recorded *P. falciparum* infections, 8 (57%) had antibodies to at least one *P. falciparum* antigen. This is consistent with results of previous studies [17,18,19] and may reflect an undiagnosed prior infection with *P. falciparum* or cross-reactivity of antibodies to the two parasite species [17,20,21,22].

As shown in Fig. 1I–1L, most of the subjects who were seropositive at recruitment and who were tested again 12 months later remained seropositive. The individual data for the rural subjects who were seropositive for the different malaria antigens at the time of recruitment and had blood collected at more than one time point are shown in Fig. 1O–1R. At an individual level, titres of antibodies against PfAMA-1, PfMSP-1<sub>19</sub> and PfMSP-2 were significantly correlated with titres of anti-PfSE Abs (data not shown). No correlations between age and the antibody titres against individual malaria antigens were found (data not shown).

#### Antibody responses to P. vivax antigens

We also investigated the antibody responses to *P. vivax* antigens, PvAMA-1, PvMSP1<sub>19</sub> and PvDBP by ELISA, all of which are P. vivax blood stage antigens and are key vaccine candidates. City naïve subjects were seronegative to all P. vivax antigens (Fig. 2A-2C). Among Rural 1 subjects, none were seropositive to PvAMA-1 (Fig. 2A), one (3%) had a borderline positive titre to PvMSP-1<sub>19</sub> (Fig. 2B) and two (6.7%) were seropositive to PvDBP (Fig. 2C). Of the Rural 2 subjects, 5 (11%), 5 (11%) and 3 (6.5%) were seropositive to PvAMA-1, PvMSP-1<sub>19</sub> and PvDBP, respectively. Overall, 8 (17.4%) Rural 2 subjects were seropositive to one or more P. vivax recombinant antigens. Of the 20 subjects known to have been previously infected with P. vivax (P. vivax only or both P. vivax and P. falciparum), 5 (25%) were seropositive to one or more P. vivax antigens and 15 (75%) were seronegative. The proportion of subjects who were seropositive to P. vivax antigens did not differ significantly between the Rural 1 and Rural 2 groups. Similarly, no significant differences in anti-P. vivax antibody titres were observed between the Rural 1 and Rural 2 groups, although the power of this analysis was poor due to the very low numbers of seropositive subjects. Antibody responses to P. vivax antigens were not different between subjects known to have been infected with P. falciparum- and those known to have been infected with P. vivax (data not shown). Most of the subjects who were seropositive to P. vivax antigens at the time of recruitment and who had samples collected at more than one time point remained seropositive over the course of the study and there was no evidence of declining titres (Fig. 2D-2F).

#### Avidity of antibodies to PfAMA-1 and PfMSP-1<sub>19</sub>

Antibody avidity tends to increase over time as a result of somatic mutation in the immunoglobulin-encoding genes of germinal centre B cells and in response to increasing competition



**Figure 1. Antibody responses to** *P. falciparum* **antigens and tetanus toxoid.** Antibody titres against tetanus toxoid (A) and *P. falciparum* antigens (B–F) among City naïve (*circle*), Rural 1 (*triangle*) or Rural 2 (*inverted triangle*) subjects at the time of recruitment were determined by indirect ELISA. Each symbol represents the antibody titre of one individual. Solid lines show the median antibody titres in each group. The Mann Whitney U test was used to analyse differences in the levels of antibodies or memory B cells among groups. Figures G–L show the percentages of all rural (i.e. Rural 1 plus Rural 2) subjects who had antibody titres above the cut-off for each antigen at the time of recruitment and 12 months later. Fischer's exact test was used to analyse differences in the proportion of seropositives at recruitment compared to 12 months later but no significant differences were observed. The antibody titres for each seropositive subject over the 12 months of the study are shown in figures M–R. Dotted lines show cut-off values calculated from a mixture model as described in materials and methods. doi:10.1371/journal.ppat.1000770.g001



**Figure 2. Antibody responses to** *P. vivax* **antigens.** A–C show antibody titres against *P. vivax* antigens among City naïve (*circle*), Rural 1 (*triangle*) or Rural 2 (*inverted triangle*) subjects at the time of recruitment. Each symbol represents the antibody titre of one individual. Solid lines show the median antibody titres in each group. The titres of antibodies of seropositive rural (i.e. Rural 1 plus Rural 2) subjects at the time of recruitment and at each time point during the 12 months of study are shown in figures D–F. Dotted lines show cut-off values calculated from a mixture model as described in materials and methods. doi:10.1371/journal.ppat.1000770.g002

between B cell clones for diminishing amounts of antigen [23,24]. To determine whether the avidity of the anti-malarial antibody response changed over time, or whether antibody avidity was associated with durability of antibody responses, the avidity indices of anti-PfAMA-1 and anti-PfMSP-1<sub>19</sub> antibodies were determined for all those individuals (as defined in Figure 1) who were seropositive to one or other antigen at the time of recruitment. Overall, avidity indices for antibodies to both antigens were higher in Rural 2 group than in Rural 1 group (Fig. 3A and 3B), and this difference was statistically significant for antibodies to PfMSP-1<sub>19</sub>. However, there was no detectable change in the avidity of antibodies to either antigen in either group over the 12 months of the study (Fig. 3C–3F).

#### Longevity of antimalarial antibodies

To determine the longevity of the antimalarial antibody responses, we analysed the change in concentrations of antibodies to PfSE, PfAMA-1 and PfMSP-1<sub>19</sub> in relation to time (in months) since the last documented malaria episode (Figure 4). The half-life of the antibody response was analysed separately for each antigen using data from Rural 2 subjects who were seropositive at the time of enrolment and for whom follow-up samples were obtained (PfSE n = 14; PfAMA1 n = 8; and PfMSP-1<sub>19</sub> n = 12) in a repeated measurements analysis including multiple data points from the same subjects. Subjects known to be infected with *P. vivax* but not known to have been infected with *P. falciparum* were not included in this analysis in order to ensure specificity to *P. falciparum*.

Mixed-effects regression models revealed very low rates of decline (converted to years) in anti- PfSE, PfAMA-1 and PfMSP-1<sub>19</sub> antibody concentrations over time and statistically, these rates could not be distinguished from zero (Table 2). The best estimates of half-lives were: 5.5 years for PfSE, 10.4 years for PfAMA-1 and 7.6 years for PfMSP-1<sub>19</sub>, respectively but, in each case, the 95% CI included infinity. Pooled regression analysis of data for antibodies to PfAMA-1 and PfMSP-1<sub>19</sub> also yielded a rate of decline that was not statistically significant from zero. Inclusion of anti-PfSE antibody data in the pooled regression analysis resulted in a marginally significant rate of decline equivalent to a half-life of 6.4 years (95% CI = 3.22, 650.48; p = 0.048). These analyses suggest that antibody responses to malaria are stably maintained in this population.

# B cell memory responses to *P. falciparum* and *P. vivax* antigens

We next enumerated memory B cells to malaria antigens and to TT using a highly sensitive ELISPOT protocol [25]. The number of subjects available for analysis was limited by availability of cryopreserved PBMCs. Antigen-specific memory B cell frequencies are presented as a percentage of the total number of IgG-secreting cells. Frequencies of TT-specific memory B cells were similar among the three study groups (Fig. 51). No spots were detected for any individual when cells were tested against the irrelevant control protein (keyhole limpet hemocyanin) and no malaria-specific spots were observed in samples from the City naïve group (data not shown).



**Figure 3.** Avidity indices for anti-PfAMA-1 and PfMSP-1<sub>19</sub> antibodies. Avidity indices for PfAMA-1 (A) and PfMSP-1<sub>19</sub> (B) at the time of recruitment were compared between Rural 1 and Rural 2 groups. Avidity indices for antibodies to both antigens were compared at the time of recruitment and at 12 months later for Rural 1 (C and D) and Rural 2 (E and F) subjects. Lines show the mean (95% CI) for each group. An unpaired Student's *t*-test test was used to confirm that there are no differences between indices at time of recruitment and 12 months later. doi:10.1371/journal.ppat.1000770.g003

At the time of recruitment, of the 21 Rural 1 subjects whose PBMCs were available, 3 (14.2%), 3 (14.2%), and 1 (4.8%) subjects had memory B cells specific to PfAMA-1 (Fig. 5A), PfMSP-1<sub>19</sub> (Fig. 5B), and PfMSP-2 (Fig. 5C), respectively. No memory B cells specific to PfCSP were found in the Rural 1 group (Fig. 5D). A much higher proportion of Rural 2 individuals had detectable memory B cells: of the 33 tested, 16 (48%), 11 (33%), 6 (18%) and 1 (3%) gave spots to PfAMA-1, PfMSP-1<sub>19</sub>, PfMSP-2 and PfCSP, respectively. Overall, 19 (58%) Rural 2 subjects had memory B cells to one or more *P. falciparum* recombinant antigens.

None of the 14 Rural 1 subjects tested had detectable memory B cells against PvAMA-1, and only one individual (7%) had detectable memory B cells to  $PvMSP-1_{19}$  (Fig. 5E and 5F). However, among the 26 Rural 2 individuals tested, 6 (23%) and 7 (27%) had memory B cells specific to PvAMA-1 and PvMSP-1<sub>19</sub>, respectively. Nine Rural 2 subjects (35%) had memory B cells specific to one or more *P. vivax* antigens.

### Stability of PfAMA-1- and PfMSP-1<sub>19</sub>-specific memory B cells

For the Rural 2 individuals, we then characterised the frequency of PfAMA-1- and PfMSP-1<sub>19</sub>-specific memory B cells in relation to



**Figure 4. Longevity of anti-malarial antibody responses.** The titres of antibodies specific to PfSE (A), PfAMA-1 (B) and PfMSP-1<sub>19</sub> (C) in relation to time since last clinical infection in *P. falciparum* exposed individuals (Rural 2 only) were determined by analyzing longitudinal data with a mixed-effects model. Each symbol represents the antibody level at each time point of one individual. The regression analysis was adjusted for inclusion of multiple data points from the same individual. Solid lines represent best fit regression lines estimating the rates of decline of antibody concentrations over time and the dashed lines represent the 95% Cl. Horizontal dotted lines indicate the cut-off as defined in Materials and Methods and Figure 1. doi:10.1371/journal.ppat.1000770.g004

time since their last documented malaria infection, using mixedeffects regression analysis (allowing for repeated measurements from individual subjects) as described above. We found that PfAMA-1- and PfMSP-1<sub>19</sub> specific memory B cells were stably maintained over time (Fig. 5G and 5H). The best estimate of the rate of change in AMA-1-specific memory B cell numbers indicated no decline during follow-up, whereas the best estimate for the half-life of MSP-1<sub>19</sub>-specific memory B cells was 10 years (Table 2). Single and pooled regression analysis of data resulted in rates of decline that, statistically, could not be distinguished from zero. Similar observations were made for memory B cells to TT (data not shown). These results indicate that memory B cell responses to malaria antigens are stably maintained in this very low transmission area.

# Correlation between antibody titres and memory B cell frequencies

It was immediately evident from the TT data that circulating memory B cells could be detected in many ( $\sim$ 47%) seronegative individuals (Figure 6G). We therefore carried out a systematic analysis of the association between circulating memory B cells and plasma antibody titres at the individual level.

Table 2. Longevity of antibody and memory B cell responses to P. falciparum antigens\*.

l													
ANTIBODY													
Antigen	No. of subjects	No. of observations	Annual decline in log titre (95% CI)	Antibody half-life (95% CI)	p-value								
PfSE	14	41	-0.1264 (-0.2809, 0.0282)	5.49 (2.47, ∞)	0.109								
PfAMA1	8	23	-0.0669 (-0.2009, 0.0671)	10.36 (3.45, ∞)	0.328								
PfMSP1 <sub>19</sub>	12	37	-0.0916 (-0.1959, 0.0126)	7.56 (3.54, ∞)	0.085								
PfMSP1 + PfAMA1	14	60	-0.0426 (-0.1672, 0.082)	16.27 (4.15, ∞)	0.503								
All	17	101	-0.1082 (-0.2153, -0.0011)	6.41 (3.22, 650.48)	0.048								
B CELL MEMORY													
Antigen	No. of subjects	No. of observations	Annual decline in log titre (95% CI)	Antibody half-life (95% CI)	p-value								
PfAMA1	13	26	0.0194 (-0.3858, 0.4245)	∞ (1.8, ∞)	0.925								
PfMSP1 <sub>19</sub>	10	19	-0.0689 (-0.5872, 0.4494)	10.06 (1.18, ∞)	0.794								
Both	15	45	-0.0928 (-0.3959, 0.2102)	7.47 (1.75, ∞)	0.548								

\*Results from a multilevel model allowing for random patient effects.

doi:10.1371/journal.ppat.1000770.t002

No correlation was observed between specific antibody titres and frequencies of memory B cells (data not shown). Among the 33 Rural 2 subjects for whom we had both antibody data and memory B cell responses to *P. falciparum* antigens at the time of recruitment, 7 (21%), 8 (24%) 2 (6%) and 0 (0%) had both circulating antibody and memory B cells to PfAMA-1, PfMSP-1<sub>19</sub>, PfMSP-2 and PfCSP respectively (Fig. 6A–6D). Four (12%), 7 (21%), 2 (6%) and 3 (9%) individuals were seropositive to the respective antigens but no B cell spots were observed whereas 9 (27%), 3 (9%), 4 (12%), and 1 (3%) were seronegative but had memory B cells to PfAMA-1, PfMSP-1<sub>19</sub>, PfMSP-2 and PfCSP respectively.

Of the 26 Rural 2 subjects for whom we had data on both antibody and memory B cell responses against *P. vivax* antigens at the time of recruitment, 1 (4%) and 2 (8%) individuals had both antibody and memory B cells against PvAMA-1 and PvMSP-1<sub>19</sub> respectively (Fig. 6E and 6F), 3 (12%) and 2 (8%) gave positive results in ELISA but not in ELISPOT, and 5 (19%) and 5 (19%) had memory B cells but not antibodies to PvAMA-1 and PvMSP-1<sub>19</sub>, respectively. These results suggest that serum antibody levels alone or memory B cell frequencies alone may not fully represent the humoral immune response to malaria parasites.

Different individuals had different patterns of antibody and memory B cell responses to the various malaria antigens. Table 3 shows the heterogeneity of such responses in all Rural 2 subjects at recruitment. In a number of cases where antigen-specific antibodies were detected, the frequencies of memory B cells were below the limit of detection (Subjects 16, 17 and 29). Likewise, in several subjects where antigen-specific antibodies were not detected, stable frequencies of antigen-specific memory B cells were observed (e.g. Subjects 27, 31 and 33).

# Discussion

Antibodies are critical in protection against blood stage malaria infection through numerous, diverse mechanisms [2,3]. In murine malaria infections, B cells are required not only for the production of protective Abs, but also for the development of T cell helper function [26]. However, the development and persistence of B cell memory responses following malaria infection has repeatedly been called into question [27]. It is widely perceived that antibody titres rapidly decline in the absence of re-infection or when individuals leave an endemic area. It is notable that most of the studies reporting short-lived antibody responses have been conducted at or following the time of acute infection, and these infections were terminated by effective antimalarial drug therapy and that these were often observations in children [13-16,28]. It is not clear, from these studies, whether the rapid decline of antibody concentrations observed in children is related to removal of antigen by chemotherapy, by consumption of antibodies and formation of antigen-antibody complexes during parasite clearance or due to limitations in the ability of the bone marrow compartment to support differentiation and/or survival of plasma cells [29,30]. However, the results of a recently published study in healthy, Gambian children in which antibody titres were found to decline more slowly both in older children and in children with persistent asymptomatic malaria infection suggests that both antigen persistence and immunologic maturity may be important in determining the longevity of the serum antibody responses [15].

One explanation for these observations might be that short-lived antibody responses are the result of induction of short-lived, but not long-lived, plasma cells following acute malaria infection. In murine models of malaria infection, primary *P. chabaudi* infection leads to expansion of short-lived, immature B220<sup>+</sup> splenic plasma cells however secondary infection is accompanied by apparently normal emergence of a larger population of fully mature (Ig<sup>hi</sup>, CD138<sup>hi</sup>, CD9<sup>+</sup>, B220<sup>-</sup>), terminally-differentiated B220- plasma cells in the bone marrow [31], indicating that memory B cells are efficiently induced by primary infection and are fully able to differentiate into long-lived plasma cells on secondary exposure to antigen. Similar studies have not been reported, to date, in humans but we were able to take advantage of a very particular epidemiological situation in rural Northern Thailand to examine the natural history of the anti-malarial B cell memory response.

In our study area, both *P. falciparum* and *P. vivax* are endemic but transmission is kept at extremely low levels by an assiduous malaria surveillance and control programme in which all detected infections are recorded and effectively treated [32]. We have thus been able to recruit a cohort of individuals whose malaria infection history over the previous 6 years are known in considerable detail and have been able to follow these individuals for a period of 12 months to observe both long- and short-term changes in their adaptive immune response to malaria. Furthermore, we were able to recruit a cohort of individuals from the same community with



**Figure 5. B cell memory responses to malaria antigens and tetanus toxoid.** B cell memory responses to *P. falciparum* antigens (A–D), *P. vivax* antigens (E and F) and tetanus toxoid (I) at the time of recruitment were determined by ELISPOT assay and are presented as the percentage of all IgG-secreting cells that are specific for each malaria antigen. Each symbol represents the memory B cell numbers for one individual. The longevity of the memory B cell responses specific to PfAMA-1 (G) and PfMSP-1<sub>19</sub> (H) were determined by analyzing longitudinal data with a mixed-effects model. Solid lines represent best fit regression lines estimating the rates of decline of memory B cell numbers over time and the dashed lines represent the 95% CI. doi:10.1371/journal.ppat.1000770.g005



**Figure 6. Correlation between ELISA and ELISPOT responses for each antigen.** For malaria antigens, data are shown for Rural 2 subjects: 33 tested against *P. falciparum* antigens (A–D) and 26 tested against *P. vivax* antigens (E and F). For TT (G), data are shown for all subjects (City naive, Rural 1 and Rural 2) whose PBMC were available (n=67). The number (and percentages) of subjects who were double positive (top left), ELISA positive but ELISPOT negative (bottom left), ELISA negative but ELISPOT positive (top right), or double negative (bottom right) are shown. doi:10.1371/journal.ppat.1000770.g006

no evidence of malaria infection in the past 6 years , and a cohort of known malaria naives from the city of Chiang Mai, where malaria transmission was eliminated more than 30 years ago (Suwonkerd W; The Ministry of Public Health; personal communication). The very low levels of malaria transmission reported in the Muang Na area [33] are confirmed by our finding that none of the study subjects were found to be infected with malaria parasites (detectable by blood film or PCR) at any point during the study, none of them showed any clinical signs of malaria infection and only one individual showed boosting of antibody responses (and against only 1 malaria antigen) during the study. In addition, anti-malarial antibody responses were not correlated with age, indicating that there is likely to be little or no effective acquired immunity to malaria in this population.

Nevertheless, some of the rural village residents with no record or recollection of malaria infection in the past 6 years (Rural 1) appeared to have experienced malaria infections at some time in their lives as shown by seropositivity to malaria antigens in ELISA and positive B cell ELISPOTs. Given the lack of evidence for acquired protective immunity in this population, and since we found no evidence of asymptomatic malaria infections, it is unlikely that these individuals had experienced undiagnosed malaria infections and thus the presence of antibodies and B cell memory responses in this group suggests that anti-malarial seropositivity can be maintained for many years in the absence of reinfection.

Overall, the prevalence and magnitude of antimalarial antibody and memory B cell responses compared favourably with the antitetanus responses. Although the frequencies of tetanus-specific memory B cells tended to be somewhat higher than the frequencies of malaria-specific memory B cells, the prevalence of antibodies to the P. falciparum schizont extract, PfMSP-119 and PfAMA-1 was in fact higher than for tetanus. Thus, despite the fact that the anti-tetanus response is likely induced by a very potent vaccine and boosted by environmental exposure or revaccination (which is routinely given during pregnancy), humoral immune responses to tetanus do not appear to be particularly more robust than those induced by infrequent natural exposure to malaria. Moreover, frequencies of malaria-specific memory B cells in Thai adults were similar to frequencies of diphtheria-specific memory B cells in UK adults (J. Palomero-Gorrindo and J. Hafalla; unpublished data). Therefore, frequencies of malaria-specific memory B cells seem to be of the same order of magnitude as responses to commonly used vaccine antigens [34,35]. Of note, Buisman et al [35] also found rather higher frequencies of memory B cells to tetanus toxoid than to other antigens.

As the time since last detected malaria infection was known for the Rural 2 group we were able to obtain estimates of the rate of Table 3. Patterns of antibody and memory B cell responses to malaria antigens and tetanus toxoid in 46 Rural 2 subjects at recruitment.

Subject	PfSE <sup>[a]</sup>	P.falcipa	<i>rum</i> antigen	s						P.vivax antigens				Tetanus toxoid	
		PfAMA1		PfMSP1 <sub>19</sub>		PfMSP2		PfCSP		PvAMA1		PvMSP1 <sub>19</sub>			
	ELISA	ELISA <sup>[b]</sup>	ELISPOT <sup>[c]</sup>	ELISA	ELISPOT	ELISA	ELISPOT	ELISA	ELISPOT	ELISA	ELISPOT	ELISA	ELISPOT	ELISA	ELISPOT
Tested ag	ainst bot	h <i>P.falcipa</i>	arum and P.	vivax aı	ntigens										
1	x	x		x	х	x									x
2	x		x	x						х		x	x	х	x
3				х	х								x		
4	х														
5															x
6	х										х		х	х	х
7															x
8															х
9															
10		х	х	х											
11	x	х		х											
12	х												х		
13	х	x	х	x							x				
14	х	х		х	х					х					
15			х											х	
16	х			х		х									
17		х													
18										х	х	х	х		х
19															x
20	х	х	х	х	х				х		х		х		х
21														х	x
22	х	х	х	х		х	х	х		х		х			
ELISPOT t	ested aga	ainst <i>P. fa</i>	<i>lciparum</i> ant	tigens o	only										
23	х		х	х	х	х	х	х			N/D		N/D	х	
24				х	х						N/D		N/D	х	
25											N/D		N/D		
26	x	x									N/D		N/D	х	
27	х	х			х			х			N/D		N/D	х	х
28			x		x		x				N/D		N/D	х	
29	Х			x							N/D		N/D		
30	х	x		x							N/D		N/D		
31	Х	х		х	х						N/D		N/D	х	
32			x		x		х				N/D		N/D		
33	Х	х	X	х	х		X				N/D		N/D		
34			x				х				N/D		N/D	x	x
35	- ا- مهم	linet 0									N/D		N/D		
	ested aga	ainst <i>P. vi</i>	wax antigen:	s only											
30	Х		N/D		N/D	X	N/D								
37			N/D		N/D		N/D		N/D		х		х		
38	X		N/D	x	N/D		N/D		N/D						x
39 Teateril (	X	X	N/D	x	N/U		IN/D		N/D		x	x		x	
rested tor	antibod	ies only													
40		×													
41	x	x		x	N/D	x		x		х		х		х	
42			N/D	х	N/D		N/U		N/D		N/D		N/D		N/D

Subject	PfSE <sup>[a]</sup>	P.falciparum antigens									P.vivax antigens				Tetanus toxoid	
		PfAMA1		PfMSP1 <sub>19</sub>		PfMSP2		PfCSP		PvAMA1		PvMSP1 <sub>19</sub>				
	ELISA	ELISA <sup>[b]</sup>	ELISPOT <sup>[c]</sup>	ELISA	ELISPOT	ELISA	ELISPOT	ELISA	ELISPOT	ELISA	ELISPOT	ELISA	ELISPOT	ELISA	ELISPOT	
43	х	х	N/D	х	N/D		N/D		N/D		N/D		N/D		N/D	
44			N/D		N/D		N/D		N/D		N/D		N/D		N/D	
45	х	x	N/D	х	N/D	x	N/D		N/D		N/D		N/D		N/D	
46	x		N/D		N/D		N/D		N/D		N/D		N/D		N/D	

<sup>[a]</sup>x indicates positive response to each antigen.

<sup>[b]</sup>Cut-off values were determined from a mixture model.

<sup>[c]</sup>A positive response was defined if the average number of spots from triplicate wells being greater than at least two times the average number of spots from the media negative control.

N/D = Not done.

doi:10.1371/journal.ppat.1000770.t003

decline of both serum antibodies and circulating memory B cells. Importantly, and in marked contrast to previously published data from African children [14], there was no evidence of any significant decline in either antibody titres or memory B cell responses to PfSE, PfAMA-1 or PfMSP-1<sub>19</sub> over periods of more than 5 years since the last known malaria infection. One reason for this discrepancy may be that previous studies have characterised the decay of the antibody response in the first few days or weeks after resolution of an acute malaria infection [14,15] which is likely to capture the initial very rapid decay in antibody titre associated with contraction of the pool of short-lived plasma cells whereas in this study, where the most recent malaria infection occurred many months or years ago, we are capturing the long term "maintenance" phase of the antibody response [36].

Our best estimates of the half-life of this maintenance phase of antibody responses to malaria antigens ranged from  $\sim 5$  to  $\sim 16$ years, putting them in the same range as the half-lives recently estimated for nonreplicating antigens such as tetanus or diptheria toxoids [37]. However, the 95% confidence intervals for the halflives of these responses all included infinity, suggesting that antibody responses to malaria may in fact be much more stable than those to nonreplicating antigens and may be maintained in a manner that is more similar to that of antiviral responses [37]. Similar estimates were obtained for the half lives of malariaspecific memory B cell responses - indicating that the circulating memory B cell pool is extremely stable - which is consistent with data from the P. chabaudi mouse model that malaria infection induces long-lived antibody responses as well as memory B cells [31]. However, whilst these analyses are consistent with a long half life for antimalarial antibody and memory cell responses, given the very wide confidence intervals around the our estimates, data from a larger cohort of study subjects is required to obtain definitive half lives for these responses.

However, it is important to note that 11 subjects (24%) who were known to have been infected with malaria had no detectable circulating memory B cells and/or antibodies and less than 50% of subjects tested had either antibodies or circulating memory B cells to PfMSP-2, PfCSP, PvAMA-1 or PvMSP-1<sub>19</sub>. Given that everyone in the study had experienced their most recent malaria infection at least 4 months before recruitment and that the genotypes of their infecting parasites are unknown, we cannot tell whether these seronegative individuals had completely failed to make a humoral response to malaria, whether they had made antibodies to polymorphic epitopes that did not cross-react with the antigens used in our assays or whether they had developed only very short-lived responses. In a previous study, in a higher malaria transmission area, very short-lived antibody responses to malaria were particularly associated with younger individuals who (presumably) had had the fewest number of malaria infections [15] suggesting that the long-lived responses seen in many of our study subjects may develop only after they have experienced a number of malaria infections. Nevertheless, given the very low levels of malaria transmission in our study area, the number of infections required to develop long-live antibody responses is likely to be quite small.

Since the pharmacological half-life of a human IgG molecule is around 21 days [38], long term maintenance of IgG titres indicates either ongoing secretion of antibodies from plasma cells or memory B cell differentiation in response to inflammatory stimuli; there is still no clear consensus on whether persisting specific antigen is required for this process [39] or not [40,41]. During inflammation, IFN-y induces plasmablasts to express the chemokine receptor CXCR3, promoting their migration into inflamed tissues [42] and thereby maximising antibody production at sites of infection. Resolution of inflammation leads to loss of survival signals, and these short-lived plasma cells die in situ. However, in the absence of prolonged antigenic stimulation, plasmablasts express another chemokine receptor CXCR4 allowing them to migrate to the bone marrow [43]. It is possible, therefore, that short term fluctuations in serum antibody concentrations may occur in response to infection with memory B cells being stimulated to differentiate into short-lived plasma cells, secrete immunoglobulins and then die. However, in the circumstances of lack of frequent re-exposure to malaria infection (such as in this study), a larger proportion of plasmablasts may differentiate into long-lived plasma cells which maintain the level of antibodies over time. Such a scenario does, of course, beg the question as what would happen to long-lived plasma cells and memory B cells under conditions of repeated or persistent malaria infection.

Immediately after malaria immunisation in humans the frequency of antigen-specific memory B cells is positively correlated with antibody titres [44], suggesting (not surprisingly) that induction of antibody secreting cells and memory B cells is linked. However, the lack of correlation that we observed between antibody titres and memory B cell responses months or years after exposure to malaria antigens confirms recent findings for anti-viral responses [37] and is consistent with accumulating experimental data from animals [45,46] and data from therapeutic B cell depletion in humans [47,48] showing that depletion of circulating memory B cells does not affect antibody titres, at least in the short-

term. Collectively, these data indicate that although both longlived plasma cells and memory B cells can be stably maintained the two populations are independently regulated and that activation of circulating memory B cells may not be required for maintenance of serum antibody titres.

High affinity antibodies are expected to play an important role in the humoral immune response. The avidity indices of antibodies against PfAMA-1 and PfMSP-1<sub>19</sub> did not change during the 12 months of study; this is not surprising since there was no evidence of reinfection of any of the subjects during the follow-up period which might have driven further avidity maturation. However the avidity of anti-PfMSP-1<sub>19</sub> antibodies was significantly higher among Rural 2 subjects than among Rural 1 individuals, supporting the notion that the Rural 2 population had had more frequent exposure to malaria parasites than the Rural 1 group.

In summary, we conclude that B cell memory responses to malaria are effectively induced and maintained - in a significant proportion of individuals - in areas of low malaria transmission. (This is, of course, an entirely separate issue from whether these particular antibodies confer protective immunity to malaria; whilst there is strong evidence to suggest that malaria-immune individuals have very effective antimalarial antibody responses [49] the antigenic targets of protective antibodies are still very poorly defined. Whilst it is possible that the subjects in this study with long-lived humoral responses to malaria antigens might be protected from reinfection, this issue was not directly addressed in this study). Although it remains possible that persistent and repeated malaria infections in areas of very high endemicity may eventually lead to B cell anergy or clonal exhaustion [50], the fact that individuals in these areas develop high titres of antimalarial antibodies and become resistant to high density malaria infections and clinical symptoms argues against this as a major impediment to the development of effective immune responses. Finally, our results are highly encouraging for vaccine developers since they imply that - once induced - anti-malarial immune responses are likely to be long-lived even in the absence of frequent boosting.

#### **Materials and Methods**

#### Study area and subjects

Study subjects were either long-term adult residents of Muang Na, a village in a low malaria transmission area in the Chiang Dao region of northern Thailand, near the border with Myanmar, or were permanent adult residents of the city of Chiang Mai where malaria transmission does not occur. Ethical approval for the study was obtained from the Research Institute for Health Sciences, Chiang Mai University, from the Ministry of Public Health, Thailand and from the London School of Hygiene and Tropical Medicine, UK. Written informed consent was obtained prior to enrolment in the study.

Subjects were interviewed to ascertain their previous malaria exposure. Residents of Chiang Mai were selected on the basis that they had not travelled to, or lived in, malaria endemic areas. In Muang Na, dates and species (*P. falciparum*, *P. vivax* or both) of malaria infections were confirmed from the records of the Office of Vector Borne Disease Control in the Department of Communicable Diseases Control at the Ministry of Public Health, which maintains detailed records of all malaria cases detected by active or passive case detection and during periodic population surveys as described in detail elsewhere [51].

Venous blood was collected in acid citrate dextrose on the day of recruitment, and again 3 months later for City naïve subjects and 3, 6 and 12 months after recruitment for rural subjects. Giemsa-stained blood films were examined for the presence of malaria parasites. Blood samples from each subject were checked for subpatent malaria parasitaemia by PCR. DNA was isolated using FlexiGene DNA extraction kits (Qiagen<sup>®</sup>) according to the manufacturer's protocol and subjected to nested PCR for *P. falciparum* and *P. vivax* as described previously [52].

As HIV infection may have an effect on immunological parameters, all subjects were tested for HIV infection (presence of anti-HIV antibodies by gel particle agglutination assay) at the time of recruitment and at the end of the study (3 months after recruitment for city subjects and 12 months after recruitment for other groups); subjects received pre- and post-test counselling from trained HIV counsellors and HIV-infected individuals were given access to the National Antiretroviral Programme. Data from HIVinfected subjects were excluded from the analysis.

#### Antigens

P. falciparum circumsporozoite protein (PfCSP) [(NANP)<sub>4</sub>] and P. falciparum merozoite surface protein-2 (PfMSP-2) were a gift from J.E. Tongren (Centre for Disease Control and Prevention, Atlanta, GA, USA). The 19kDa fragments of P. falciparum and P. vivax MSP-1 (PfMSP-1<sub>19</sub> and PvMSP-1<sub>19</sub>) were gifts from A. Holder (National Institute of Medical Research, London, UK) and the proteins were expressed as described [53]. P. falciparum apical membrane antigen-1 (PfAMA-1) was a gift from R.F. Anders (LaTrobe University, Victoria, Australia); the equivalent *P.vivax* antigen (PvAMA-1) was a gift from B. Farber and A. Thomas (Biomedical Primate Research Centre, Rijswik, Netherlands). P. vivax duffy binding protein (PvDBP) was a gift from L.H. Carvalho Centro de Pesquisas René Rachou, Fundação Oswaldo Cruz, Belo Horizonte, MG, Brazil). Since Thai populations are routinely vaccinated with tetanus toxoid (TT), antibody responses to TT were included as a positive control. TT was obtained from the National Institute of Biological Standards and Control (Health Protection Agency, Hertfordshire, UK). Keyhole limpet haemocyanin (KLH) was from Thermos Fisher Scientific (Northumberland, UK).

Continuous cultures of *P. falciparum* (3D7 strain were maintained in the laboratory [54] and were periodically shown to be free from Mycoplasma contamination by polymerase chain reaction (PCR) (Venor<sup>®</sup> GeM, Minerva Biolabs). Mature schizonts were obtained by gradient centrifugation over 60% Percoll (Amersham Biosciences), adjusted to a concentration  $1 \times 10^8$  schizont-infected red blood cells (iRBC)/ml and exposed to three freeze/thaw cycles to obtain *P. falciparum* schizont extract (PfSE).

#### Enzyme-linked immunosorbent assay (ELISA)

Plasma antibody levels were detected by indirect ELISA, as described [55]. Briefly, Immulon 4HB (Dynatech) or Maxisorb (Nunc) plates were coated with antigen (at a concentration equivalent to 10<sup>5</sup> iRBC/ml for PfSE, 0.5 µg/ml for *Plasmodium*derived antigens and TT) in bicarbonate buffer (pH 9.6) overnight at 4°C. Plates were blocked with PBS containing 1% non-fat milk powder. Diluted plasma samples (1:200 for PfCSP, 1:1000 for PfSE, PfMSP-1<sub>19</sub>, PfMSP-2, PvMSP-1<sub>19</sub> and PvDBP; 1:2000 for PfAMA-1, PvAMA-1 and TT) were incubated in dubplicate. Plates were subsequently developed with anti-human IgG horseradish peroxidase conjugate (Caltag Laboratories, Invitrogen, Paisley, UK) followed by *o*-phenylenediamine substrate (Sigma). The enzyme reaction was terminated with sulphuric acid (2N) and absorbance was then read at 492 nm on a Spectra MR plate reader (Dynex Technology).

Antibody levels were determined by comparison to a standard curve (derived by serial dilution of a pool of hyperimmune plasma collected in The Gambia, which was given an arbitrary value of 1,000 units/ml of anti-PfSE Abs) on each plate, as described previously [12].

# Preparation of peripheral blood mononuclear cells (PBMCs)

PBMCs were separated from citrated blood by gradient centrifugation over Ficoll-Hypaque (Amersham Biosciences). Contaminating erythrocytes were removed by incubating with lysis buffer (0.15 M NH<sub>4</sub>Cl, 10 mM KHCO<sub>3</sub>, 0.1 mM Na<sub>2</sub>EDTA) at RT for 5 minutes. The cells were washed twice with RPMI, resuspended in 10% human AB serum/RPMI (R10 culture medium), counted, adjusted to the required concentration and cryopreserved in 10% dimethylsulfoxide (DMSO)/foetal calf serum.

# Stimulation of cryopreserved PBMC for B cell ELISPOT assay

Cryopreserved PBMCs were quick thawed in a 37°C water bath. The cells were washed twice with warm RPMI, resuspended in R10 culture medium and added at a concentration of  $1 \times 10^6$ cells/ml to a 24 well culture plate. The cells were stimulated with medium alone or with a mixture of *Phytolacca americana* pokeweed mitogen (1/100,000 dilution; a gift from M. Causland and S. Crotty, La Jolla Institute of Allergy and Immunology, CA, USA), 6 µg/ml CpG 2006 (Qiagen/Operon), and 1/10,000 dilution of Staphylococcus Aureus Cowan (SAC) (Sigma), as previously described [25]. The culture plates were incubated in 5% CO<sub>2</sub> at 37°C for 5 days.

### B cell ELISPOT assay

B cell ELISPOT assays were performed as described previously [25]. Briefly, ELISPOT plates (Millipore) were coated with donkey anti-human IgG (H+L) (Jackson ImmunoResearch), or with 1 µg/ ml recombinant malaria proteins overnight at 4°C. After washing once with PBS-T and three times with PBS, 200 µl of 1% bovine serum albumin in RPMI were added to each well and incubated for 2 hours at 37°C, 5% CO<sub>2</sub>. Cultured PBMCs were recovered from the 24 well culture plates, washed, transferred directly to antigen-coated ELISPOT plates and incubated for 6 hours at 37°C, 5% CO<sub>2</sub>. After 4 washes with PBS and 4 washes with PBS-T, 100 µl of Biotin-SP-conjugated donkey anti-human IgG (Jackson Immunoresearch) were added to each well and the plates were incubated overnight at 4°C. The plates were washed, 100 µl of alkaline phosphatase-streptavidin (Vector Laboratories) was added and incubated for one hour at room temperature. After three washes with PBS-T and three washes with PBS, 100 µl of 5bromo-4-chloro-3-indolyl phosphate/ nitro blue tetrazolium alkaline phosphatase substrate solution (Vector Laboratories) were added to each well and the reaction was allowed to proceed for 8 minutes before being stopped with distilled water. In vitro restimulated PBMCs incubated overnight with an irrelevant protein, KLH, as well as PBMCs cultured without stimulation and then incubated overnight with malaria antigens were used as negative controls. Since no malaria-specific spots were detected in city naïve individuals, this group was not be used to set a cut-off for positivity. Rather, a positive ELISPOT response was defined when spots were observed in 2 or more replicate wells and where the total number of spots in the antigen-coated wells was at least twice the number observed in the negative control wells.

## Avidity assay

An enzyme immunoassay for determination of antibodies against malaria antigens was carried out as described above.

### References

Following the incubation step of sera with antigens, one duplicate set of sera was treated with 4.0 M guanidine dissociating solution (Guanidine Hydrochloride, Sigma) for 10 minutes prior to washing with PBS-T. Avidity indices were calculated as the ratio of the OD of guanidine -treated wells to the OD of the untreated wells.

### Statistical analysis

To determine whether an individual was seropositive for a particular antigen (PfSE, Pf- or Pv-derived antigens, or TT), cutoffs for positive antibody titres were calculated using a mixture model, which assumes that untransformed titres for seropositive and seronegative samples each follow a normal Gaussian distribution [56,57]. Mann Whitney U test was used to analyse differences in the levels of antibodies or memory B cells among groups (GraphPad Prism software). Fischer's exact test was used to analyse differences in the proportion of positive individuals between Rural 1 and Rural 2 groups, as well as differences in the proportion of seropositives at recruitment compared to 12 months later. Decay rates for antibody titres and memory B cell frequencies were calculated using logarithmically transformed data from subjects who were seropositive or memory B cell positive, respectively, at recruitment. The effect of time since malaria infection was analysed using a log-linear mixed-effects regression model incorporating Gaussian random intercepts. This resulted in an estimate of the decay rate of antibody titres or memory B cell frequencies, assuming a single-exponential decay model. Half-lives were calculated from the estimated decay rate and the boundaries at 95% confidence interval obtained from the mixed-effects model. Where the decay rate is a positive value, the calculated half-life is reported as infinity. All analyses were undertaken using Stata (version 10, Statacorp LP).

# **Supporting Information**

**Figure S1** Antibody responses against PfSE in *P. falciparum* (*square*) and *P. vivax* (*diamond*) exposed subjects. Each symbol represents the antibody titre of one individual. Dotted lines show cut-off values calculated from a mixture model as described in materials and methods. Solid lines show the median antibody titres in each group.

Found at: doi:10.1371/journal.ppat.1000770.s001 (0.03 MB DOC)

# Acknowledgments

We thank all the subjects of this study for their participation. We would like to thank Tanapol Panya and staff from the Vector Borne Disease Control Centre, Chiang Mai for their help with the field work; the HIV counselling team from the Research Institute for Health Sciences; Chanida Rattanapinyopong and Tunlaya Ruankham for technical assistance. We are grateful to Dr. Wannapa Suwonkerd for very helpful discussion and to M. Causland and S. Crotty, La Jolla Institute of Allergy and Immunology, USA for discussions.

### **Author Contributions**

Conceived and designed the experiments: JW EMR JCRH. Performed the experiments: JW CS JCRH. Analyzed the data: JW LCO JC PHC KT EMR JCRH. Contributed reagents/materials/analysis tools: JW CS JC PHC WL JCRH. Wrote the paper: JW EMR JCRH.

 Gupta S, Snow RW, Donnelly CA, Marsh K, Newbold C (1999) Immunity to noncerebral severe malaria is acquired after one or two infections. Nat Med 5: 340–343.

Wipasa J, Elliott S, Xu H, Good MF (2002) Immunity to asexual blood stage malaria and vaccine approaches. Immunol Cell Biol 80: 401–414.

- Beeson JG, Osier FH, Engwerda CR (2008) Recent insights into humoral and cellular immune responses against malaria. Trends Parasitol 24: 578–584.
- Langhorne J, Ndungu FM, Sponaas AM, Marsh K (2008) Immunity to malaria: more questions than answers. Nat Immunol 9: 725–732.
- Struik SS, Riley EM (2004) Does malaria suffer from lack of memory? Immunol Rev 201: 268–290.
- Stephens R, Albano FR, Quin S, Pascal BJ, Harrison V, et al. (2005) Malariaspecific transgenic CD4(+) T cells protect immunodeficient mice from lethal infection and demonstrate requirement for a protective threshold of antibody production for parasite clearance. Blood 106: 1676–1684.
- D'Imperio Lima MR, Alvarez JM, Furtado GC, Kipnis TL, Coutinho A, et al. (1996) Ig-isotype patterns of primary and secondary B cell responses to *Plasmodium chabaudi chabaudi* correlate with IFN-gamma and IL-4 cytokine production with CD45RB expression by CD4+ spleen cells. Scand J Immunol 43: 263–270.
- Wykes MN, Zhou YH, Liu XQ, Good MF (2005) *Plasmodium yoelii* can ablate vaccine-induced long-term protection in mice. J Immunol 175: 2510–2516.
- Carvalho LJ, Ferreira-da-Cruz MF, Daniel-Ribeiro CT, Pelajo-Machado M, Lenzi HL (2007) Germinal center architecture disturbance during *Plasmodium berghei* ANKA infection in CBA mice. Malar J 6: 59.
- Taylor RR, Egan A, McGuinness D, Jepson A, Adair R, et al. (1996) Selective recognition of malaria antigens by human serum antibodies is not genetically determined but demonstrates some features of clonal imprinting. Int Immunol 8: 905–915.
- Udhayakumar V, Kariuki S, Kolczack M, Girma M, Roberts JM, et al. (2001) Longitudinal study of natural immune responses to the *Plasmodium falciparum* apical membrane antigen (AMA-1) in a holoendemic region of malaria in western Kenya: Asembo Bay Cohort Project VIII. Am J Trop Med Hyg 65: 100–107.
- Drakeley CJ, Corran PH, Coleman PG, Tongren JE, McDonald SL, et al. (2005) Estimating medium- and long-term trends in malaria transmission by using serological markers of malaria exposure. Proc Natl Acad Sci U S A 102: 5108–5113.
- Cavanagh DR, Elhassan IM, Roper C, Robinson VJ, Giha H, et al. (1998) A longitudinal study of type-specific antibody responses to *Plasmodium falciparum* merozoite surface protein-1 in an area of unstable malaria in Sudan. J Immunol 161: 347–359.
- Dorfman JR, Bejon P, Ndungu FM, Langhorne J, Kortok MM, et al. (2005) B cell memory to 3 *Plasmodium falciparum* blood-stage antigens in a malaria-endemic area. J Infect Dis 191: 1623–1630.
- Akpogheneta OJ, Duah NO, Tetteh KK, Dunyo S, Lanar DE, et al. (2008) Duration of naturally acquired antibody responses to blood-stage *Plasmodium falciparum* is age dependent and antigen specific. Infect Immun 76: 1748–1755.
- Asito AS, Moormann AM, Kiprotich C, Ng'ang'a ZW, Ploutz-Snyder R, et al. (2008) Alterations on peripheral B cell subsets following an acute uncomplicated clinical malaria infection in children. Malar J 7: 238.
- Kumar N, Folgar JP, Lubega P (1992) Recognition of *Plasmodium falciparum* asexual stage antigens by antibodies in sera from people exposed to Plasmodium vivax. Am J Trop Med Hyg 47: 422–428.
- Jangpatarapongsa K, Sirichaisinthop J, Sattabongkot J, Cui L, Montgomery SM, et al. (2006) Memory T cells protect against *Plasmodium vivax* infection. Microbes Infect 8: 680–686.
- Zevering Y, Khamboonruang C, Rungruengthanakit K, Tungviboonchai L, Ruengpipattanapan J, et al. (1994) Life-spans of human T-cell responses to determinants from the circumsporozoite proteins of *Plasmodium falciparum* and *Plasmodium vivax*. Proc Natl Acad Sci U S A 91: 6118–6122.
- Igonet S, Vulliez-Le Normand B, Faure G, Riottot MM, Kocken CH, et al. (2007) Cross-reactivity studies of an anti-*Plasmodium vivax* apical membrane antigen 1 monoclonal antibody: binding and structural characterisation. J Mol Biol 366: 1523–1537.
- Bracho C, Perez H (1992) An immunocytochemical test for the diagnosis of antibodies to *Plasmodium vivax*. Parasite Immunol 14: 481–487.
- Kim YM, Hwang HA, Yun WS, Kim SI, Lee KW, et al. (2004) Efficacy of the merozoite surface protein 1 of *Plasmodium vivax* as an antigen for ELISA to diagnose malaria. Yonsei Med J 45: 129–134.
- Phan TG, Paus D, Chan TD, Turner ML, Nutt SL, et al. (2006) High affinity germinal center B cells are actively selected into the plasma cell compartment. J Exp Med 203: 2419–2424.
- Adams CL, Macleod MK, James Milner-White E, Aitken R, Garside P, et al. (2003) Complete analysis of the B-cell response to a protein antigen, from in vivo germinal centre formation to 3-D modelling of affinity maturation. Immunology 108: 274–287.
- Crotty S, Felgner P, Davies H, Glidewell J, Villarreal L, et al. (2003) Cutting edge: long-term B cell memory in humans after smallpox vaccination. J Immunol 171: 4969–4973.
- Langhorne J, Cross C, Seixas E, Li C, von der Weid T (1998) A role for B cells in the development of T cell helper function in a malaria infection in mice. Proc Natl Acad Sci U S A 95: 1730–1734.
- Pierce SK, Miller LH (2009) World Malaria Day 2009: what malaria knows about the immune system that immunologists still do not. J Immunol 182: 5171–5177.
- Kinyanjui SM, Bull P, Newbold CI, Marsh K (2003) Kinetics of antibody responses to *Plasmodium falciparum*-infected erythrocyte variant surface antigens. J Infect Dis 187: 667–674.

- Pihlgren M, Friedli M, Tougne C, Rochat AF, Lambert PH, et al. (2006) Reduced ability of neonatal and early-life bone marrow stromal cells to support plasmablast survival. J Immunol 176: 165–172.
- Pihlgren M, Schallert N, Tougne C, Bozzotti P, Kovarik J, et al. (2001) Delayed and deficient establishment of the long-term bone marrow plasma cell pool during early life. Eur J Immunol 31: 939–946.
- Stephens R, Ndungu FM, Langhorne J (2009) Germinal centre and marginal zone B cells expand quickly in a second *Plasmodium chabaudi* malaria infection producing mature plasma cells. Parasite Immunol 31: 20–31.
- 32. The Ministry of Public Health, Thailand (2003) Manual for Malaria Control; Bureau of Vector Borne Diseases, editor. Bangkok: The Agricultural Cooperative Federation of Thailand., Limited.
- Department of Disease Control, The Minstry of Public Health. Report on malaria cases. http://203.157.41.98/malaria/freereoprt/rep.php?free = 1& PHPSESSID = 368625f72ed8fbe6801903db82d4c152.
- 34. Clutterbuck EA, Oh S, Hamaluba M, Westcar S, Beverley PC, et al. (2008) Serotype-specific and age-dependent generation of pneumococcal polysaccharide-specific memory B-cell and antibody responses to immunization with a pneumococcal conjugate vaccine. Clin Vaccine Immunol 15: 182–193.
- Buisman AM, de Rond CG, Ozturk K, Ten Hulscher HI, van Binnendijk RS (2009) Long-term presence of memory B-cells specific for different vaccine components. Vaccine 28: 179–86.
- Manz RA, Hauser AE, Hiepe F, Radbruch A (2005) Maintenance of serum antibody levels. Annu Rev Immunol 23: 367–386.
- 37. Amanna IJ, Carlson NE, Slifka MK (2007) Duration of humoral immunity to common viral and vaccine antigens. N Engl J Med 357: 1903–1915.
- Hopkins RJ, Kramer WG, Blackwelder WC, Ashtekar M, Hague L, et al. (2004) Safety and pharmacokinetic evaluation of intravenous vaccinia immune globulin in healthy volunteers. Clin Infect Dis 39: 759–766.
- Benson MJ, Elgueta R, Schpero W, Molloy M, Zhang W, et al. (2009) Distinction of the memory B cell response to cognate antigen versus bystander inflammatory signals. J Exp Med 206: 2013–25.
- Bernasconi NL, Traggiai E, Lanzavecchia A (2002) Maintenance of serological memory by polyclonal activation of human memory B cells. Science 298: 2199–2202.
- Montes CL, Acosta-Rodriguez EV, Merino MC, Bermejo DA, Gruppi A (2007) Polyclonal B cell activation in infections: infectious agents' devilry or defense mechanism of the host? J Leukoc Biol 82: 1027–1032.
- Muchlinghaus G, Cigliano L, Huchn S, Peddinghaus A, Leyendeckers H, et al. (2005) Regulation of CXCR3 and CXCR4 expression during terminal differentiation of memory B cells into plasma cells. Blood 105: 3965–3971.
- Tokoyoda K, Egawa T, Sugiyama T, Choi BI, Nagasawa T (2004) Cellular niches controlling B lymphocyte behavior within bone marrow during development. Immunity 20: 707–718.
- Crompton PD, Mircetic M, Weiss G, Baughman A, Huang CY, et al. (2009) The TLR9 ligand CpG promotes the acquisition of *Plasmodium falciparum*-specific memory B cells in malaria-naive individuals. J Immunol 182: 3318–3326.
- 45. Alwayn IP, Xu Y, Basker M, Wu C, Buhler L, et al. (2001) Effects of specific anti-B and/or anti-plasma cell immunotherapy on antibody production in baboons: depletion of CD20- and CD22-positive B cells does not result in significantly decreased production of anti-alphaGal antibody. Xenotransplantation 8: 157–171.
- DiLillo DJ, Hamaguchi Y, Ueda Y, Yang K, Uchida J, et al. (2008) Maintenance of long-lived plasma cells and serological memory despite mature and memory B cell depletion during CD20 immunotherapy in mice. J Immunol 180: 361–371.
- Anolik JH, Barnard J, Cappione A, Pugh-Bernard AE, Felgar RE, et al. (2004) Rituximab improves peripheral B cell abnormalities in human systemic lupus erythematosus. Arthritis Rheum 50: 3580–3590.
- Vallerskog T, Gunnarsson I, Widhe M, Risselada A, Klareskog L, et al. (2007) Treatment with rituximab affects both the cellular and the humoral arm of the immune system in patients with SLE. Clin Immunol 122: 62–74.
- 49. Bouharoun-Tayoun H, Attanath P, Sabchareon A, Chongsuphajaisiddhi T, Druilhe P (1990) Antibodies that protect humans against *Plasmodium falciparum* blood stages do not on their own inhibit parasite growth and invasion in vitro, but act in cooperation with monocytes. J Exp Med 172: 1633–1641.
- Pierce SK (2009) Understanding B cell activation: from single molecule tracking, through Tolls, to stalking memory in malaria. Immunol Res 43: 85–97.
- Thimasarn K, Jatapadma S, Vijaykadga S, Sirichaisinthop J, Wongsrichanalai C (1995) Epidemiology of Malaria in Thailand. J Travel Med 2: 59–65.
- Snounou G, Viriyakosol S, Zhu XP, Jarra W, Pinheiro L, et al. (1993) High sensitivity of detection of human malaria parasites by the use of nested polymerase chain reaction. Mol Biochem Parasitol 61: 315–320.
- Burghaus PA, Holder AA (1994) Expression of the 19-kilodalton carboxyterminal fragment of the *Plasmodium falciparum* merozoite surface protein-1 in Escherichia coli as a correctly folded protein. Mol Biochem Parasitol 64: 165–169.
- Trager W, Jensen JB (1976) Human malaria parasites in continuous culture. Science 193: 673–675.
- 55. Egan AF, Chappel JA, Burghaus PA, Morris JS, McBride JS, et al. (1995) Serum antibodies from malaria-exposed people recognize conserved epitopes formed by the two epidermal growth factor motifs of MSP1(19), the carboxy-terminal fragment of the major merozoite surface protein of *Plasmodium falciparum*. Infect Immun 63: 456–466.

- Vyse AJ, Gay NJ, Hesketh LM, Pebody R, Morgan-Capner P, et al. (2006) Interpreting serological surveys using mixture models: the seroepidemiology of measles, mumps and rubella in England and Wales at the beginning of the 21st century. Epidemiol Infect 134: 1303–1312.
- 57. Ohuma EO, Okiro EA, Bett A, Abwao J, Were S, et al. (2009) Evaluation of a measles vaccine campaign by oral-fluid surveys in a rural Kenyan district: interpretation of antibody prevalence data using mixture models. Epidemiol Infect 137: 227–233.